Author’s response to reviews

Title: Anti-inflammatory effects of a methanol extract of Dictamnus dasycarpus Turcz. root bark on imiquimod-induced psoriasis

Authors:

Minjee Choi (chl6789@naver.com)
Si-Yong Kim (pau26@naver.com)
Jung Ryu (ryuuuuu316@nate.com)
Jinhee Lee (ljh0041@nate.com)
Wookbong Kwon (6funny@naver.com)
Soyoung Jang (wkdthdus21@naver.com)
Dongjun Kim (kdj1012@naver.com)
JunKoo Yi (79lee38@korea.kr)
Hyungwoo Kim (kronos7@pusan.ac.kr)
Sung Kim (shkim92@knu.ac.kr)
Myoung Kim (ok4325@knu.ac.kr)
Zae Young Ryoo (jaewoong64@hanmail.net)

Version: 4 Date: 19 Nov 2019

Author’s response to reviews:

We carefully reviewed Editor’s opinion and revised the manuscript.

Our response to Editor’s comments

1. Please clarify in the 'Ethics Approval and Consent to Participate' section if any permissions were necessary to collect the specimens in your study. Collection of specimens must comply with institutional, national, or international guidelines. Field studies should be conducted in accordance with local legislation, and the manuscript should include a statement specifying the appropriate permissions and/or licences.
In response to your grateful comments, we have modified the content of the 'Ethics Approval and Consent to Participate'.

2. Please provide details in your Methods section on who undertook the formal identification of the plant material used in your study. Please also confirm whether a voucher specimen of this material has been deposited in a publicly available herbarium, and include this information in your manuscript. A deposition number should be included, if available.

- In response to your opinion, we have added an explanation for formal identification and deposition information (Page 6 line 16-21).

3. In your ARRIVE checklist we note that Item 7, points B, C and D are not "not applicable". Please can you ensure you have provided details regarding these items in your manuscript. We also note that Item 8B, 10B, 18B, 18C and 19 need revising.

- In response to your thoughtful comments, we have modified the content of the ARRIVE checklist.

4. Please put your responses to the reviewers'/editors' comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

- Thank you for your comment.